Author's response to reviews

Title: Validation of a brief suicide risk screening tool in a Chinese population

Authors:

Chia-Yi Wu (jennycyw@ntu.edu.tw)
Hui-Chun Huang (aihch@yahoo.com.tw)
Shu-I Wu (shui624@yahoo.com.tw)
Fan-Ju Sun (jisun.b612@ms1.mmh.org.tw)
Chiu-Ron Huang (joan0727@xuite.net)
Shen-Ing Liu (maryliuyip@gmail.com)

Version: 3
Date: 26 December 2013

Author's response to reviews: see over
December 26th, 2013

Re: Validation of a Brief Suicide Risk Screening Tool in a Chinese Population

Dear Chief Editor

According to the Reviewers opinions, we have made point-to-point revisions and amended our manuscript concurrently. Most of these opinions are useful to improve the quality of our paper. Please see the attached files for more details. We believe that the revised manuscript is valuable in providing evidence-based results in the validation of the Chinese SAD PERSONS Scale. We would be happy if this article can be successfully published in your journal, and we look forward to hear from you soon.

Yours sincerely

Shen-Ing Liu
MD, PhD
Department of Psychiatry, Mackay Memorial Hospital
45 Min-sheng Road, Tam-Shui, New Taipei City, Taiwan
Telephone: 00-886-28094661 ext 3055